Author’s response to reviews

Title: Growth, immune and viral responses in HIV infected African children receiving highly active antiretroviral therapy: a Growth, immune and viral responses in HIV infected African children receiving highly active antiretroviral therapy: a prospective cohort study

Authors:

Philippa M Musoke (pmusoke@mujhu.org)
Peter Mudiope (pmudiope@mujhu.org)
Linda Barlow-Mosha (lbarlow@mujhu.org)
Patrick Ajuna (pajuna@mujhu.org)
Danstan Bagenda (dbagenda@mujhu.org)
Michael Mubiru (mmubiru@mujhu.org)
Thorkild Tyelleskar (Thorkild.Tyelleskar@cih.uib.no)
Mary Glenn Fowler (mgfowler@mujhu.org)

Version: 3 Date: 26 March 2010

Author’s response to reviews:

Melissa Norton MD
Editor in Chief
BMC Pediatrics

Re: Growth, immune and viral responses in HIV infected African children receiving highly active antiretroviral therapy: a prospective cohort study
Manuscript ID 7493930337133912

Dear

I would like to thank you for giving me the opportunity to make the correction to the files I uploaded. The only change to the manuscript file was that the previous manuscript did not have the tables at the end of the manuscript, but only had the pages of table legends and titles. I also reduced the number of words in the abstract to 352 by using abbreviations of BAZ, WAZ, and HAZ in the results section but maintaining the full spelling in the methods section of the abstract. I also separated figure 1 and 2 from each other so they could be uploaded separately.

Thank you

Sincerely,

Dr. Philippa Musoke
Associate Professor